Author's response to reviews

Title: Bilateral Hilar Lymphadenopathy in a Young Female: Case Report Seema Varma, MD; Shilpi Gupta, MD; Raymond ElSouidie, MD; Meekoo Dhar, MD; Neville Mobarakai, MD Sanford R. Nalitt Institute of Cancer and Blood Related Diseases, Staten Island University Hospital, New York.

Authors:

Seema Varma MD (svarma@siuh.edu)
Shilpi Gupta MD (sgupta@siuh.edu)
Raymond El-Souidie MD (relsouidie@siuh.edu)
Meekoo Dhar MD (mdhar@siuh.edu)
Neville Mobarakai MD (nmobarakai@siuh.edu)

Version: 3 Date: 27 March 2007

Author's response to reviews:

This is the first case of Bronchiolitis-Obliterans Organizing Pneumonia (BOOP) presenting with extensive hilar and mediastinal lymphadenopathy. The clinical presentation and radiologic features were typical for sarcoidosis.

BOOP can clinically mimic any disease, however, hilar or mediastinal lymphadenopathy is not included in the wide spectrum of radiologic findings associated with BOOP.